journal homepage: www.ojs.sabauni.net

Saba Journal Of Information Technology And Networking (SJITN)



i



All Rights Reserved© 2021 Saba Journal of Information Technology and Networking, Published by Saba University.

journal homepage: www.ojs.sabauni.net

Saba Journal Of Information Technology And Networking (SJITN)



EDITOR IN CHIEF Ibrahim Ahmed Al-Baltah

ADVISORY BOARD

Dr. G. Radhamani, India Dr. Nidhal K. El-Abbadi, Iraq Dr. Emad Abu-Shanab, Jordan **Prof. Gerald Robert Midgley, UK** Dr. Tawfiq S. Barhoom, Palestine Dr. Wesam Bhaya, Iraq Dr. Ahmad M. Aznaveh, Iran Dr. Mohamed M. Elammari, Libya Dr. Hisham Abushama, Sudan Dr. Ali Al-Dahoud, Jordan **Dr. Mohammad Ibraheem, Egypt** Dr. Ahlal H. Montaser, Libya Dr. Safaa Ahmed Hussein, Egypt Dr. Maha Ahmed Ibrahim, Egypt Dr. Basem Mohamed Elomda, Egypt Dr. Alaa El-din Mohamed Riad, Egypt Dr. Rehab Fayez Sayed, Egypt

Dr. Iyad M. Al-Agha, Palestine Dr. Enas Hamood, Iraq Dr. Assad norry, Iraq Dr. Amjad Farooq, Pakistan Dr. Sanjeev Gangwar, India Dr. Waleed Al-Sitt, Jordan Dr. N.Sudha Bhuvaneswari, India Dr. Ali . Al-Sharafi, Saudi Arabia Dr. Essam Said Hanandeh, Jordan Dr. Ramadan Elaiess, Libya Dr. Izzeldin M. Osman, Sudan Dr. Rasha Osaman, United Kingdom Dr. Eiman Kanjo, Saudi Arabia Dr. Huda Dardary, USA

All Rights Reserved© 2021 Saba Journal of Information Technology and Networking, Published by Saba University.

journal homepage: www.ojs.sabauni.net



iii

Saba Journal Of Information Technology And Networking (SJITN)

Table of Contents

Title	Page no.
Yemeni Riyal Exchange Rate Prediction Using Predictive Models Based on	1-8
Artificial Intelligent	
Mahammad Hasham Almanniah Ahmad Yourof Sacad Danham Mahammad	
Monammea Hashem Almourish, Anmea Tousoj Saeea, Bornan Monammea Radman Abdulfattah Famail Ra Almui Dalil Abdulhari Al aarshi	
Kaaman, Abauijanan Esman ba Alawi, Dani Abaubari Ai-qershi	
Foreign currency, such as the dollar, plays a fundamental role in controlling market	
prices in many countries, because the import and export process takes place through	
these foreign currencies. Predicting the exchange rates of these changing currencies is	
a task with great challenges. The main purpose of this paper is to build a system based	
on Artificial Intelligence to predict the exchange rate between US Dollar (USD) and	
Yemen Riyal (YER) for the next day or several days in the future. The artificial	
intelligence exploit two techniques, Recurrent Neural Network (RNN) and Machine	
Learning (ML). The proposed model uses recurrent Neural Network that exploits long	
short-term memory (LSTM) and machine learning with three models (Linear	
Regression Model, Random Forest Regression and Gradient boosted regression) to	
predict the currency USD/YER for the next day or several days in the future with the	
highest accuracy. The objective function of training and testing the prediction models	
to find out performance of proposed models by calculating the root mean square error	
which came out to be very low. The best results were obtained using Random Forest	
Regression model in root mean square error training that reached 0.0448 and the best	
result in root mean square error testing was obtained using Linear Regression model	
that reached 0.1070.	0.25
A Review of 101 in Education: Benefits and Challenges	9-25
Ruqaih Hussein Salman and Ibrahim Ahmed Al-Baltah	
Internet of Things (IoT) has emerged as a new technology with the aim of improving	
diverse sectors. Due to the ubiquity of IoT devices, academic institutions and schools	
are seeking to incorporate IoT into education activities. The goal of this paper is to	
review the current state-of-the-art in order to explore the revolution that has been	

occurred since the incorporation of IoT in education. Besides, the possible opportunities that could be gained for students, instructors, and admins from the	
adaption of IoT in the educational process are classified.	
Apracadabra application platform for intelligent and immediate-services based IoT technology	26-33
Al-Marhabi Zaid Ali, Al-Hamdi Ayeda G, Habeb Abduljlil Abduljlil	
Apracadabra application platform visions are to connect multi types of technologies: RFID, sensors, actuators, person's users etc., the opportunity of bring out the IoT technology into reality became possible at the present through the integration of many available technologies, especially in the under development countries (not high infrastructure). The IoT provides many important features, which transformed the internet from communicating with people to the possibility communicating between	
things "smart devices" to do queries, surveillance and other smart services.	
Due to high cost on attaching a sensors or RFID card in some services or providers we propose Apracadabra, Smart Services to contribute on developing societies, would be unable to jump suddenly to the accurate IoT technology concept.	
This paper address Apracadabra application platform to be the right solution for under development countries to provide same services as IoT in terms of speed, we can call it the first generation of the IoT. Apracadabra based on attaching an account number or	
IP address for each user (server - clients), is the services provided by this user will be presented to other interests with the possibility of evaluating those services by the beneficiaries, this platform gives the user's location and all available ways of	
Apracadabra Smart Services (ASS) name came from aladeen Arabic story (Shobic	
Lubic (شببك ليك) or Apracadabra) which means helping others or clients to get	
whatever they need immediately, it is a mobile devices application running on Android	
system due to android system widespread and characterized by many features, also we	
can access Apracadabra through any web browsers.	

journal homepage: <u>www..ois.sabauni.net</u>



Article

Yemeni Riyal Exchange Rate Prediction Using Predictive Models Based on Artificial Intelligent

Mohammed Hashem Almourish¹, Ahmed Yousof Saeed², Borhan Mohammed Radman², Abdulfattah Esmail Ba Alawi², Dalil Abdulbari Al-qershi³

¹Department of Communication & Computer Engineering - Taiz University - Yemen.
 ²Department of Software Engineering - Taiz University - Yemen.
 ³Department of computer Sciences, Faculty of Applied Sciences, Taiz university, Taiz, Yemen.

Article info

Article history:

Accepted: March. 2021

Keywords:

USD/YER; RNN; LSTM; ML; Linear Regression Model; Random Forest Regression; Gradient Boosted Regression.

Abstract

Foreign currency, such as the dollar, plays a fundamental role in controlling market prices in many countries, because the import and export process takes place through these foreign currencies. Predicting the exchange rates of these changing currencies is a task with great challenges. The main purpose of this paper is to build a system based on Artificial Intelligence to predict the exchange rate between US Dollar (USD) and Yemen Riyal (YER) for the next day or several days in the future. The artificial intelligence exploit two techniques, Recurrent Neural Network (RNN) and Machine Learning (ML). The proposed model uses recurrent Neural Network that exploits long shortterm memory (LSTM) and machine learning with three models (Linear Regression Model, Random Forest Regression and Gradient boosted regression) to predict the currency USD/YER for the next day or several days in the future with the highest accuracy. The objective function of training and testing the prediction models to find out performance of proposed models by calculating the root mean square error which came out to be very low. The best results were obtained using Random Forest Regression model in root mean square error training that reached 0.0448 and the best result in root mean square error testing was obtained using Linear Regression model that reached 0.1070.

* Corresponding author: Mohammed Hashem Almourish E-mail: Mohmedalmourish@gmail.com

1. Introduction

Since the last few decades, the foreign exchange markets in the world have witnessed an increase in all parts of the world. The exchange rate is an important parameter and crucial parameter in businesses and companies which has made forecasting of exchange rates an important area in all business sectors. A small change in the exchange rate affects an increase in the prices of the products and this forces the seller to pay more money, which creates problems and a burden on the citizen. Therefore, the economic stability of countries is important in in the stability of the exchange, as economically foreign stable countries have a more valuable currency. And therefore the exchange rate becomes easier to predict, while unstable countries have a less valuable currency, and the currency becomes more difficult to predict. Investors around the world choose economically stable countries to build their companies and deposit their money in their banks in their own currencies. In recent years, the exchange rate in Yemen has become unstable due to economic instability, and this has led to fluctuations in exchange rates, which made exchange predicting rates very difficult. Therefore, in this paper, we decided to build a system for predicting exchange rates between Dollar and Yemen Riyal. There are systems that are used to forecast exchange rates, but they are complicated and their accuracy is weak, which makes prediction less reliable and therefore it was necessary to find more efficient and reliable systems. In recent times, the use of artificial intelligence in many sectors has become very important, especially in the economic sector, such as predicting exchange rates and stock prices in companies or banks. This paper presents a system to predict the exchange rate between the Dollar and Yemen Riyal using artificial intelligence that uses Recurrent Neural Network and Machine

Learning due to the high accuracy of prediction. We tested all models on the dataset and found that the best model is linear regression model that has minimum root mean square error in testing prediction that reached 0.1070.

The paper is organized into six section: The second section presents the background material and previous work which is most related to the field of this research. In section 3 the methodology is presented. Proposed techniques and how they're implemented are given in section 4. The experiment results of the proposed method presented in section 5. In section 6, conclusions and the future work of the research are presented.

2. Related Work

We present an overview of literature that relates to work presented here. The change in currency exchange rates sometimes leads to a large loss in trade operations in the event that it was not properly predicted [1]. So, one of the most direction of research in economic sciences is exchange rates prediction using artificial intelligence. Jefferson et al. [2] presented a paper for Canadian-US exchange rate prediction using different artificial intelligence models and applied a comparison between the performance of these models. Kamruzzaman et al. [3] presented three models of neural networks that predict the Australian exchange rate against six foreign currencies. These neural networks showed an ability to predict exchange rates closely. Reference [4] proposed Turkish prediction rate using different neural network models. Weigend et al. proposed prediction models for the German exchange rate against the dollar based on neural network and stochastic walking model, and the results for neural networks were better [5].

Mehreen R et al. [6] presented a paper for exchange rate predictions which made a novel approach to predict exchange rate by using CGP

and RNN called (RCGPANN). Podding et al. [7] presented the results of the German exchange rate prediction model against the dollar, and the results obtained were compared with the regression analysis model. In [8], authors propose a system that making comparison between two different model (ANN and RNN) which used to predict exchange rates Rupees/ US, Rupees/ Euro and Rupees/ Pound Sterling. The paper by [9] presented the results of a prediction comparison between the US dollar and the German dollar using neural networks and linear models. Fernando et al. [10] presented a paper that compared the efficiency of various artificial neural network to stock prices prediction. Multilayer perceptron has been built with Levenberg Marquardt, GRU and LSTM for stock market prediction. In this paper, prediction model of the exchange rates between Dollar against Yemen Riyal has been built using recurrent neural network that uses LSTM model and machine learning with three models (Linear Regression Model, Random Forest Regression and Gradient boosted regression).

3. Methodology

In this section, we will discuss the method that used to build a system for predicting exchange rates between Dollar against Yemen Riyal by using three main phases approach. The following steps summarize the methodology used to achieve the proposed method:

3.1. Dataset Collection

The dataset was obtained from the website [11]. The dataset contains the USD/YER exchange from date 16-jun-2016 to 31-jun-2020. This data consists of 1077 days of data for exchange rates are as listed in the Table 1 and shown in Figure 1.

|--|

Date	Rate
16-jun-2016	250.1
17-jun-2016	250.15
20-jun-2016	249.935



Figure 1. Plot of currency exchange rate value for USD _ YER.

3.2. Training Phase

After preparing data, we start training the model on the collected dataset with the following steps:

- Load Dataset: In this step, dataset is loaded to fit the training phase.
- **Data preprocessing:** This step is aimed to normlize the dataset between zero and one.
- -Training Using LSTM model: In this step the system was trained by the LSTM model and the dataset was divided into 80% for training and 20% for testing. Figure 2 shows the proposed model for predicting exchange rates using the LSTM model.
 - Training Using Machine Learning Models: In this step the system was trained using Machine Learning Models (Linear Regression Model, Random Forest Regression and Gradient boosted regression). The dataset was divided into 80% for training and 20% for testing. Figure 2 shows the proposed models for

predicting exchange rates using the using Machine Learning Models.



Figure 2. The proposed model for predicting exchange rates using the LSTM model.



Figure 3. The proposed model for predicting exchange rates using the using Machine Learning Models.

3.3. Testing Phase

At this phase, future data is entered for the system to make predictions based on the training model that was done. The testing phase consists of following steps:

- **Data pre-processing:** This step is aimed to normalize the dataset between zero and one.
- **Prediction phase:** Prediction phase is the last phase, in which all models (LSTM, Linear Regression Model, Random Forest Regression and Gradient boosted regression) produce accurate prediction of USD/YER exchange rate for the next day.

4. Techniques

In this section, we will discuss proposed techniques that used to build an effective system that able to predict the exchange rates between Dollar against Yemen Riyal for the next day.

4.1. Artificial Neural Network

Artificial neural networks are part of artificial intelligence that is used to simulate the human brain and has many practical applications, including pattern recognition and also in the fields of prediction [12]. ANN contains a collection of nodes called neurons which connected together by link called edge as shown in Figure 4. This edge used to transmit the signal between neurons. Each edge has weight value which increase or decrease in self-learning process. ANN has three layers: input, hidden and output layer.

Input layer: Data will input to the neural network by using input layer.

Hidden layer: These layers used to process a data that come from input layer.

Output layer: After processing the data in the previous layer, it is passed on to the output layer.

Each input in the input layer has value denoted by Xn. and every edge in neural network has weight value denoted by Wn.

The value of the output layer denoted by Yn which calculate as the following equation 1:

$$\begin{split} Y &= Activation_function(\sum_{j=1}^{n} Xj * Wij) \quad (1) \\ \text{The activation function as see in rule1 used to} \\ determine if the value will pass (1) or not (0). \end{split}$$



Figure 4. Architecture of ANN.

4.2. Recurrent Neural Network

A RNN is a class of neural networks that differs from other types by contain internal memory which used to store the inputs that come from each node (see Figure 5). This difference making the RNN more precise than others. One of the limitations of recurrent neural networks is that they have short-term memory. This memory was not enough if there are long series of information. So, LSTM built to solve this problem.



Figure 5. Architecture of Recurrent Neural Network.

- Long Short Term Memory: LSTM is specific type of RNN that introduced by Hochreiter et al. [13] which used to learn the long-term dependency. It is containing cell state, input gates, output gates and forget gates as shown in Figure 6.



Figure 6. LSTM gates.

- **Input gate.** This gateway is used to discover the value of the input used to update the memory and combines the previous output with the new input and passes it through the sigmoid activation function. An input gate has an equation of the following form:

 $it = \sigma(Wi. [ht - 1. Xt] + bi)$ (2)

Were (σ) is the a logistic sigmoid function, *Wi* and *bi* are the parameter to learn, *Xt* is the input sequence, *ht* is the hidden state of the encoder at time t. A candidate layer gate has an equation of the following form:

 $C't = \tanh(Wc.[ht - 1.Xt] + bc$ (3)

Where *tanh* is the hyperbolic tangent function, Wc and bc are the parameter to learn, C't - this internal memory of the cell [14].

- Forget gates. This gate is used to detect details that should to remove them from block which used sigmoid activation function. If the value of output (ft) is 0 then previous output will remove and keep it if (ft) = 1 in the number of C't. A forget gates has an equation of the following form:

$$ft = \sigma(Wf. [ht - 1. Xt])$$
(4)

- Output gates. This gate determines the value of

 $ot = \sigma(Wo.[ht - 1.Xt] + bo)$ (5) ht = ot * tanh(C't) (6)

4.3. Machine Learning

Machine Learning methods are part of artificial intelligence which used for prediction. The proposed model uses three ML algorithms including, linear regression model, random forest regression and gradient boosted regression.

Linear Regression Model: The Linear Regression is a part of machine learning, that is used to fit predictive model to a data can be used to predict future data. A linear regression model looks as a line has an equation as follows:

 $\hat{y}=w[0]*x[0]+...+w[p]*x[p]+b$ (7) where x[0] to x[p] are the features, w and b are the parameter to learn and \hat{y} is the prediction model [15].

Random Forest Regression: Random forest is a part of machine learning was introduced by Breiman in 2001, by integrating between classification and regression tree [16]. The Random forest is construct by integrating the multiple decision trees randomly, that trains several trees in parallel. The results are aggregated, through model averaging as shown in Figure 7.

Gradient Boosted Regression: The gradient boosted regression is a machine learning technique, which is a grouping of trees in a sequence where each tree corrects the errors of the previous tree, making the model smaller in terms of memory and making predictions faster and thus the model becomes more powerful. Figure 8 shows the structure of Gradient boosted regression.



Figure 7. Random Forest structure.



Figure 8. Gradient boosted regression structure.

5. Results

In this paper, exchange rate prediction model has been built between US Dollar (USD) and Yemen Riyal (YER) for the next day and or several days in the future using predictive models based on Artificial Intelligence. The proposed models, namely Recurrent artificial neural network model (LSTM) and Machine learning models (Linear Regression, Random Forest Regression and Gradient boosted regression) can be used to build an effective system that able to predict the USD/YER exchange rate for the next day. Every model shows independent performance. There are

S.J.I.T.N Vol .9 No.1 (2021)

two different currencies Dollar and Yemen Rivals presented in our dataset. We have selected one dollar against Yemen Riyal. Data was analyzed using proposed models and was provided experimental evaluation of these models in Table 2. The Table 2 summarizes the results for the training sets and the prediction results in the testing sets with 100 epochs and 20 batch sizes by using proposed models (LSTM, Linear Regression, Random Forest Regression and Gradient boosted regression) to predict the exchange rate of USD/YER for the next day. As demonstrated in the below table, the best result was obtained using Random Forest Regression model in root mean square error (RMSE) of training that reached 0.0448 and the best result of prediction in root mean square error of testing was obtained using Linear Regression model that reached 0.1070.

Model	RMSE	RMSE
	training	testing
LSTM	0.2485	0.3580
Linear Regression	0.0531	0.1070
Random Forest	0.0448	0.1185
Regression		
Gradient boosted	0.0450	0.1157
regression		

The result of prediction values of training and testing can be seen in Figure 9. From the below plot we can note that the blue line denoted to the actual daily USD/YER exchange rate values, the yellow line represented the data which used as training data and the green line represented the predicted values for testing set.

6. Conclusion

The main conclusions of this paper is to predict USD /YER currency exchange rate for the next day by using predictive models based on Artificial Intelligence techniques. This paper describes tow artificial intelligence techniques, RNN and ML, and presents a comparison between LSTM and models of ML (Linear Regression, Random Forest Regression and Gradient boosted regression). The objective function of training and testing the prediction models are to minimize the error in the model. The best result with the Random Forest Regression model in RMSE of training that reached 0.0448 and the best result of prediction of testing was obtained using Linear Regression model that reached 0.1070. In the future searches, we will propose another types of currencies such as Euro and Pound Sterling to predict exchange rate between Yemen Rivals and them. Also will propose others algorithm that will give better result.



Figure 9. Prediction Performance of USD/YER currency: actual USD /YER vs predicted values for testing set.

7. References

 [1] S. Ranjit, S. Shrestha, S. Subedi and S. Shakya, "Comparison of algorithms in Foreign Exchange Rate Prediction." In Proceedings on 2018IEEE 3rd International Conference on Computing, Communication and Security (ICCCS). IEEE,2018.

- [2] J. Davis, A. Episcopos, and S. Wettimuny, "Predicting direction shifts on Canadian–US exchange rates with artificial neural networks." International Journal of Intelligent Systems in Accounting, Finance & Management, 10(2), pp: 83–96, 2001.
- [3] J. Kamruzzaman and R. Sarker. "Forecasting of currency exchange rates using ANN: A case study." In Proceedings of the 2003 International Conference on Neural Networks and Signal Processing, vol 1. IEEE, pp: 793–797, 2003.
- [4] C. Kadilar and H.Alada, "Forecasting the exchange rates series with ANN", Istanbul University Econometrics and Statistics e-Journal, vol.9, no. 1, 2009.
- [5] A. Weigend, B. Huberman and D. Rumelhart. "Predicting sunspots and exchange rates with connectionist networks, in M. Casdagli and S. Eubank eds., Nonlinear Modeling and Forecasting," Addison-Wesley, pp: 395–432, 1992.
- [6] M. Rehman, G. Khan, S. Mahmud. "Foreign currency exchange rates prediction using CGP and recurrent neural network." IERI Procedia, 10, pp: 239-244, 2014.
- [7] A. Podding. "Short term forecasting of the USD/DEM exchange rate," in Proceedings of first International Workshop on Neural Networks in Capital Market, London, 1993.
- [8] A. Aydin and S. Cavdar. "Comparison of prediction performances of artificial neural network (ANN) and vector autoregressive (VAR) Models by using the macroeconomic variables of gold prices, Borsa Istanbul (BIST) 100 index and US Dollar-Turkish Lira (USD/TRY) exchange rates." Procedia Economics and Finance 30, pp: 3-14, 2015.
- [9] T. Hann and E. Steurer, "Much ado about nothing? Exchange rate forecasting: Neural networks versus linear models using monthly

and weekly data", Neurocomputing 10, pp: 323–339,1996.

- [10] A.J.P. Samarawickrama; T.G.I. Fernando."A recurrent neural network approach in predicting daily stock prices an application to the Sri Lankan stock market". In Proceedings of the 2017 IEEE International Conference on Industrial and Information Systems (ICIIS), Peradeniya, Sri Lanka, 15–16 December 2017, pp: 1–6, 2017.
- [11] USD/YER Historical Data. (31-jun-2020).[Online]. Available: https://sa.investing.com/ currencies/usd-yer-historical-data.
- J. Frankenfield, Artificial Neural Network.
 [Online]. Available: https://www.investopedia.com/terms/a/artificia
 l-neural-networks-ann.asp, 2020.
- [13] S. Hochreiter, and J. Schmidhuber. "Long short-term memory." Neural computation, vol. 9, pp: 1735-1780, 1997.
- [14] A. Mittal. (2019) Understanding RNN and LSTM. [Online]. Available: https://towardsdatascience.com/understandingrnn-and-lstm-f7cdf6dfc14e. 2019.
- [15] A. Müller and S. Guido. "Introduction to machine learning with. Python: a guide for data scientists." O'Reilly Media, Inc., 2018.
- [16] L. Breiman, "Random forests," Machine learning, vol. 45, no. 1, pp. 5–32, 2001.

journal homepage:<u>www..ojs.sabauni.net</u>



Article

A Review of IoT in Education: Benefits and Challenges

Ruqaih Hussein Salman and Ibrahim Ahmed Al-Baltah

Information technology department, faculty of computer science and IT, Sana'a University, Sana'a, Yemen

Article info

Abstract

Article history:

Accepted: April. 2021

Keywords: Internet of things; Modern Digital Tools; Attendance Tracking Student; Student assessment Internet of Things (IoT) has emerged as a new technology with the aim of improving diverse sectors. Due to the ubiquity of IoT devices, academic institutions and schools are seeking to incorporate IoT into education activities. The goal of this paper is to review the current state-of-the-art in order to explore the revolution that has been occurred since the incorporation of IoT in education. Besides, the possible opportunities that could be gained for students, instructors, and admins from the adaption of IoT in the educational process are classified.

* Corresponding author: Ruqaih Hussein Salman E-mail: r.salman@su.edu.ye

All Rights Reserved © 2021 Saba Journal of Information Technology and Networking, Published by Saba University.

1. Introduction

The development and growth of the internet is naturally exponential. In the past 25 years, internet has continuously expanded to connect people from all over the world through computers, laptops, smart phones and other devices. Nowadays, with the development of the global internet, various devices such as household appliances, automobiles, different electrical devices and various smart devices can also use internet services to communicate, which leads to the creation of the Internet of Things (IoT). IoT enables different real-world objects called "things" to communicate with each other on the global internet using services that support Internet (IP) through wired or Protocol wireless communication networks. These things can perceive the surrounding environment and act on it autonomously, meanwhile transforming the surrounding physical world into a very large information and knowledge base [1].

The term "Internet of Things" was firstly coined by Kevin Ashton in 1999. Various terms are derived from the literature of IoT technology, including Internet of Everything (IoE), Industrial IoT, Internet of Anything, Internet of People, Internet of Services, Web of Things, Machine-to-Machine communication or Internet of Signs, Internet of Data or Internet of Processes [2].

According to [3], IoT stands for "anything at all, depending on requirements". Cisco defines IoT as a network of connected physical objects. Cisco pointed out: "IoE integrates people, processes, things and data, together to make connections of network more valuable than ever and relevant before turning information into actions that can bring new capabilities, unprecedented economic opportunities, and richer experience, to individuals, companies, and countries [4].

IoT network connects different types of devices, such as personal computers, tablets, PDAs,

laptops, smart phones and other handheld embedded devices. These devices collect useful information through various sensors and data collection technology, and then transfer it to other processing equipment for interpretation and decision-making [3].

The core function of the IoT lies on the ability of devices to exchange information without any necessary human intervention. This phenomenon is called as communication of Machine-to-Machine (M2M) [5, 6]. IoT-based systems communicate via wireless technologies [7], such as Radio Frequency Identification (RFID) [8, 9], ZigBee [10], Wireless Sensor Network (WSN) [11]. Figure 1 shows the connectivity between devices to exchange information.



Figure 1. IoT connectivity [1]

IoT applications have been used in multiple fields such as smart retail, medical services, smart home, customer service, industrial Internet and environmental monitoring. Now, due to its ubiquitous nature. schools and academic institutions are seeking to incorporate the IoT into educational activities to benefit teachers. students, employees, and the entire education system. For this purpose, many researchers conducted to suggest applications of IoT to solve various models, objectives, specializations and concepts in the field of education [12]. The main purpose of this work is to discuss the usefulness

and applications of the IoT in the field of education. Moreover, we focus on opportunities of IoT in field of education especially with all interconnected actors (instructor, students and staff) in the education process. In addition, we discuss the challenges of IoT in field of education. The rest of the paper is organized as follows: Section 2 presents the literature review; Section 3 presents the IoT in learning process, Section 4 presents the benefits of IoT; Section 5 presents the challenges of IoT; Section 6 presents the conclusion of paper.

2. Literature Review

Educational technology has played an important role in connecting and educating students. IoT technology has a significant impact on the education field. IoT does not only change the traditional learning and teaching methods, it also infrastructure of changes the educational institutions [13]. IoT technology may play an important role in improving education at all including universities, levels colleges and schools. From students to teachers, from classrooms to campuses, everything can benefit from this technology. There are many researchers discussed the IoT in education.

M. Kassab et al. [12] summarized and categorized the existing benefits and challenges of integration of the IoT with education field, and proposed a framework to appropriately position and promote new research activities. They conducted a systematic literature review to extract, evaluate, identify and synthesize some published researches about the symbiosis of IoT in education to answer some research questions and determine the state evidence with in-depth analysis.

IoT is the most challenging platform to specify the association of physical objects in the near future. M. Al-Emran et al. [14] highlighted the latest developments in the adoption of IoT applications in education and provided various challenges opportunities and for future experiments. The authors summarized the horizons of IoT technologies in education in general, medical education and vocational training in particular, in addition to discussing wearable technologies and their relationship to education and education with green IoT.

A. C. Martín et al. [15] analyzed the current situation of smart education. A study of the development of published papers on smart education, its technologies, keywords and methods of data collection and analysis, education level and localization. The developed systems in the selected papers were also reviewed, and their quality and feasibility evaluated. At the end of the research, they identified the opportunities and the most areas important for improvement. They concluded that, despite the vagueness of the term, there do exist developments today that can make educational technology more suitable for learners, thereby providing a foundation for smarter ways of learning.

A. M. Alalade et al. [7] provided a little insight on the basic knowledge of the IoT, its architecture. its fundamentals. core and furthermore contributed to understand the application of the IoT in the education field. This research was carried out in a more advanced and state-of-the-art IoT-centric teaching plan, which included smart laboratories, smart classrooms, and the entire smart campus.

D. D. Ramlowat and B. K. Pattanayak [1] addressed the importance of IoT technology in the field of education to improve efficiency of teaching and learning process. They focused on the innovations suggested by several authors about implementing IoT resources in the field of education, such as distance education, computer science education, consumer green education and

medical science education. The interrelationship between IoT and the cloud-computing paradigm discussed in education.

S. Pervez et al. [16] discussed several aspects of IoT with respect of productivity levels, which helps contemporary learners and educators to learn from the results of the processed data. They showed the different learning trends and the success rate of these application methods. They also discussed the various educational perspectives that must be incorporated when developing educational curricula to meet the requirements of society's smart literacy. The results assisted decision makers and recommended the use of more professional methods to improve the better teaching model platforms in specific communities.

Das et al. [17] studied and discussed some of IoTbased applications in depth, such as smart classroom, smart e-learning, smart library, etc. In addition, they discussed the technologies needed to enable urban and rural residents to use an education system based on the IoT. They tried to address this gap and find a new direction to help integrate the knowledge of rural and urban students and empower people through the power of IoT. Their work discussed the importance of applications and technologies required to build a smart education system, and helped researchers think about new applications and technologies based on the IoT.

3. IoT and Participates in Education Process

Now, due to IoT's ubiquitous nature, academic institutions and schools are seeking to incorporate IoT into educational activities to benefit students, lecturers, administrators, and the entire education system. Many researchers discussed IoT's opportunities in education and classified it. M. Kassab et al. [12] identified three dimensions to classify benefits of IoT in education field and discussed its benefits for each dimension such as perception, learning principles and education sitting (mode). Figure 2 shows the threedimensional scheme for IoT in field of education. A perception is a way to look at something, to understand or interact with. More specifically, perception refers to the stakeholder role, which requires the IoT system to provide one of its services in an educational environment. Stakeholders have a goal with respect to the system, one that can be satisfied through its operations. They have identified three different main stakeholders from the extracted scenarios. Each of the main stakeholders has a perception when they interact with IoT technology when deployed in the education field: student, instructor and staff(Admin) [12]. However, the authors presented IoT applications and services in education in relation to the three stakeholders involved in the educational process according to their perception in general without classifying IoT services for each participant in the educational process in a detailed manner.



Figure 2. The three-dimensional scheme for Internet of things in education[2]

IoT has the potential to greatly change the educational process and the relationship between the participants in education process. In this paper, we focus on IoT in field of education specifically, IoT opportunities for each service related to the stakeholder involved in the educational process separately. Figure 3 shows IoT opportunities for the services of the three stakeholders involved in the educational process.



Figure 3. IoT's opportunities of stakeholders involved in the educational process.

4. IoT Benefits of Student

The definition of student includes scholar, learner, and pupil. A student is one of the key chain links in the education process. We can study the opportunities of the IoT in education specifically, for students. Moreover, IoT can provide students with disabilities with the help and support they need to achieve a good quality of life and allow them to participate in social and economic life. In this section, we will discuss the most important opportunities of IoT provided from а student's perspective with/without disabilities as follows:

4.1. IoT-based Adaptive Learning

There are many opportunities that have made learning adaptive according to student's needs. IoT not only provides learning, but also supports learning. IoT may create a customized learning environment according to students' individual learning style, personality, level of achievement, strengths, feelings, interests, or cultural background of the students. Moreover, IoT helps to make learning available when needed by the student and at the time when the student needs the information or knowledge. The most important and newest opportunities based on IoT which have made learning adaptive discussed and proposed by many researchers such as the use IoT to assist students with disabilities, personal learning and ubiquitous learning.

Sula et al. [18] implemented the performance of a new system based on IoT and P2P technology for supporting learning process and improving the life quality of children with Autism Spectrum Disorder (ASD). They used SmartBox device and JXTA-Overlay platform to monitor the children and create P2P communication between children, caregivers and therapists.

R. Mehta [19] talked about five ways of IoT to change education and the field of learning from special education. Anyone who is slightest interested in the education of special children will know the role of bibs of technology. Children with autistic respond very well to tablets, as do children with other diseases. Children with ADHD, dyslexia or other learning disabilities can greatly help them through any available technology. Along with this, smart schools also have a system for educating people like special students. The public will therefore have more understanding and openness to special individuals. It is easier for special children to work in a neutral environment and alone.

Sula et al. [20] proposed an intelligent auxiliary environment system that used heuristic diagnostic teaching process. The purpose was to identify each student's mathematics learning ability and their creative characteristics. The proposed system used computers, sensors, RFID tag readers and SmartBox devices to support the learning of students with autism spectrum disorders by providing a case-by-case basis-based personalized feedback and practices.

S.J.I.T.N Vol .9 No.1 (2021)

M. C. Domingo [56] proposed an appropriate learning environment for learners with special educational needs. IoT created a user friendly environment for people with disabilities and helped for their social integration. The smart devices and applications facilitated the management and execution of daily activities made them more.

With regard to personal learning, A. Uzelac et al. [21], a smart classroom was proposed, which can measure the level of interest of students by using cameras and microphones to monitor their behavior. The influence of different parameters in the physical environment (such as temperature, environmental noise and CO2 level) on students' attention is also analyzed. This helps to customize the way of personalized learning plans.

S. Meacham et al. [22] proposed an IoT system that enabled a large number of students in lecture class and laboratories to conduct personalized education. The proposed system based on a case study based on the work of a medium-sized university in the UK.

M. Kravčík et al. [23] mainly focused on IoT and related technologies to study how to improve this filed. They claimed that proper interpretation and analysis of big educational data, which can make personal learning and training experiences more accurate, personalized and adaptable, thereby making them more effective, attractive and efficient. The information collected through a rich palette of available allows for better personalization and adaptiveness of learning.

Recently, the concept of personal learning environment (PLE) has emerged, opening new doors for more effective learning and overcoming many limitations of the traditional TEL model. According to Chatty et al. [24], they provided details of the theory, design, implementation, and evaluation of mixed and match personality learning environment framework (PLEF). The main purpose of PLEF is to help learners to use various digital media and data to create custom learning mashups.

Finally, learning can be adapted to the student's needs by ubiquitous learning which means students will able to learn from anywhere at any time. A. Labus et al. [25] focused on integrating wearable technology into the e-learning structure so that ubiquitous learning can be achieved through devices interactions and collaborative work. They proposed an integrated model that enables this goal to be achieved. The proposed model consisted of physical wearable technology infrastructure, software to run devices, and cloud computing platform.

An attempt of transferring learning from traditional form to e-learning was proposed by

K. Papadokostaki et al. [26]. They explore the experience API specification and examine how it can be used for the implementation of adaptive learning applications to make ubiquitous learning real for different learning level

R. Xue et al. [27] proposed a framework based on IoT to construct ubiquitous learning environment. This framework was divided into three main layers including application layer, network layer, and perception layer. Besides, its architecture was clearly presented, and its applicability was articulated through an example.

K. Mershad and P. Wakim [28] proposed "Remote Lectures" to enhance LMS framework through the utilization of IoT infrastructure. Communication towers were used to transmit multimedia and data to distant locations. Some IoT devices were installed in the classroom to record and transmit the delivered lectures and its related material such as lecture notes, interactive exercises to the LMS of the enrolled students. Their proposed gave students more flexibility in managing their schedules. However, the proposed feature allows students to capture all important resources related to the lecture from anywhere and at any time.

4.2. IoT-based Increase Student Engagement

Utilizing IoT infrastructure in learning increase the growth of student's engagement in the learning process. Student engagement is a multifaceted theory that constitutes of three factors; behavioral, emotional and cognitive engagement. Many researchers discussed opportunities for IoT to make education more enjoyable and attractive, which leads to maximizing student engagement in education. The most important is making the learning process more interactive, just like playing the game or providing interactive environments between students themselves and between students and teachers.

If we consider student engagement in education as the degree of attention, curiosity, interest, inquisitiveness, motivated, optimism, awareness and passion that students/learners show, which extends to the enthusiasm level they must improve and study in their education. Two-sided interactive feature of IoT device makes it an effective tool that facilitate greater student participation in the process of learning. Through various communication technologies available in the IoT learning environment, students can answer questions and provide feedback quickly and repeatedly. Also, the instructors can define various strategies and techniques for education in order to capture the feedback of learners in real time [29]. The principle of improving students' "motivation" is addressed through bridging the communication between teachers and students using the IoT. In this context, based on cloud computing platform technology, combined with internet and web technologies, an innovative interactive platform for ideological education is proposed [30]. The platform includes five functional layers and three-layer architecture. The feasibility of this platform has been verified by studio visual as the development environment and the expression of ASPX documents [30].

S. Satu et al. [31] proposed an educational platform called the intelligence of learning things (IoLT). It is a blended learning method based on the IoT, which can enhance traditional education systems through innovative and interactive learning strategies and technologies. In this platform, different applications and devices can be used to collaborate and share their ideas with stakeholders (for example, teachers and students) through the system using IoT. Therefore, the proposed model able to ensure smarter and more effective method than traditional methods.

J. Henry et al. [32] introduced a computer algorithm used to quantify student engagement based on individual feedback, attendance, and punctuality. The data algorithm uses game points to depict the results to meet the needs of gamification, serious games and edutainment. By including the IoT, they obtained behavioral engagement data related to real-world events. In addition, they introduced the IoT as a tool to measure the impact of the real-world environment on student engagement.

R. Garris et al. [33] proposed an input-process output game model, which focused on the instructional games and it engagement to learners to accomplish better learning outcomes. The proposed model detailed a repeated game cycle that include user behavior, user judgement and feedback of system and encourages interacting with a game.

Z. AjazMoharkan et al. [34] discussed some applications of IoT in education in general, and in e-learning in particular. In addition, thev proposed smart learning model based on the IoT and e-learning gamification technology. The proposed educational model was smart and participatory due to the application of gamification concepts, which made learning more attractive.

4.3. IOT Improves learning results of students IOT can assist institutions to improve the quality of learning and teaching by providing a richer learning experience and real-time actionable insights into student performance.

Bob Nilsson et al. [35] argued that today's education field has used IoT devices such as tablets, e-books, fitness bands, sensors, augmented reality and virtual reality headsets to track and monitor students' various Aspects, such as understanding students' learning patterns.

A. Uzelac et al. [21], a smart classroom was proposed, which can measure the level of students' interest by using cameras and microphones to monitor their behaviour. The influence of different parameters in the physical environment (such as temperature, environmental noise and CO2 level) on students' attention is also analyzed. This helps to customize the way of personalized learning plans.

5. IoT Benefits of Instructor

University professors or school teachers-they are all called instructors. They also get a lot of benefits from IoT in education. When we come to IoT's opportunities and their innovative technics of instructors, it is clear that technological developments have brought some noticeable improvements in the teaching process. [41]. IoT helps instructors to focus on student learning instead of wasting time to maintain large procedure when dealing with students in the classroom. Networked devices can save time and allow instructors to focus on cultivating some extra qualities of students [39]. In this section, we will discuss four of the most important IoT opportunities that can improve teaching process. The Strong Sides of IoT from an instructor perspective as follows.

5.1. IoT-based Modern Digital Tools

IoT offers modern digital tools to better explain new subjects and classroom management. A. M. Alalade et al. [36] discussed different physical applications of IoT in different areas of academia, including IoT in the classroom. Recently, a very powerful teaching platform called smart boards have been used in the classroom. With online presentations and videos, it can help teachers explain lectures more easily. In addition, interactive tools such as educational games and exercises can be used through the smart board, which is very helpful for educating students effectively. Furthermore, they discussed other ways to incorporate IoT in education involves the use of Web-based tools, programs, smart marker, and smart cameras.

J. Vihervaara and T. Alapaholuoma [37] emphasized that IoT is capable to improve vocational education and training in many ways. IoT can be an effective educational tool for building coaching skills of students trying to find answers to undiscovered problems using electronic devices.

The term "classroom management" represents the way a teacher achieves ordered in the classroom. With the help of smart devices such as cameras, microphones and many other sensors, which can measure how students are satisfied with the things they learn about the term, classroom management becomes very simple and easier to understand. Moreover, teacher can know when to talk louder, when the concentration of students is falling down and when to take a break [38]. IoT can help to improve the learning environment in the classroom for higher education. It provides a good level of control and a large amount of sensor data, allowing instructors to adjust classroom conditions according to the subject and time of day continuously. The intensity of natural light and artificial light can be controlled. It can improve air quality and noise levels. By monitoring the ambient sound level at the back of the room, you can alert instructors when their voice is difficult to understand [35].

5.2. IoT-based Attendance Tracking Student

IoT automates students' attendance tracking instead of manual process done by the instructor, which in turns save some times for instructors. Therefore, IoT-based attendance management system will reduce the overhead of calculating attendance hours [39].

One of the simple techniques that is focused on taking student attendance is proposed by F. Akhter et al. [39]. This system utilizes IoT to collect student attendance and to store them securely in database. The system architecture is constituted of some modules including FRID, NodeMCUV3, and fingerprint sensor which is used as an authentication tool. Whereas, RFID module is responsible for scanning RFID tags and sends data to the central server. The system then able to generate attendance reports based on the collected data.

When students attend the classroom, they can use smart devices (such as Nymi) to automatically record their attendance. The device is a wearable "smartband" that uses the wearer's ECG pattern to verify a student's identity [40]. On the other hand, a face tracking approach was adapted in the work proposed by S. R. Babu et al. [41] to automate students' attendance process. They used techniques such as face recognition, object detection, and convolutional neural networks. The proposed system has reduced training time and computational complexity compared to the current system.

Z. Jiang [42] proposed IoT-based student attendance management system, which employs some key technologies such as embedded development, ZigBee, RFID, and others. Using the RFID technology and WSN technology to capture the student's trajectory, it can provide record and analyze students' activities trajectory, analysis data for the student attendance management system, and grasp real-time dynamics of students.

The system can effectively strengthen the supervision of student activities, reduce the burden of attendance management of teachers and students, and improve the level of intelligent management, which has important practical significance for the construction of an intelligent campus.

P. Tan at al. [43] designed and implemented an IoT-based teaching management system to help teacher to perform automatic attendance record and strengthen student learning motivation. It used an open source hardware platform such as RFID-RC522 and Node MCU to implement a Wi-Fi-supported RFID reader. The proposed WiRF system used to record students' attendance and record their behavior automatically to assist the teacher. The system has a positive impact on student learning in higher education and improves student attendance rate.

H. Rjeib et al. [44] proposed a system of attendance management (AMS) students based on RFID, and an information service system for an academic field that used RFID technology in addition to programmable logic circuits (such as Arduino) and Web-based applications. The proposed system not only save time, but also reduce documentation work without any power consumption.

On another hand, a web-based attendance system is proposed in [45] which use NFC technology that is available in smart phones. The attendance rate will be automatically saved on the server when the student clicks the matrix card on the NFC smart phone. Both students and teachers can check their presences via their smart phone [45].

B.I Ahmad [46] proposed NFC-based smart attendance system to simplify the process of attendance, by simply the lecturer's NFC-based

S.J.I.T.N Vol .9 No.1 (2021)

mobile device in the class or touching an attendance poster. The system presented substantially improved the current attendance registration system and eliminate many paper works involved in it. Other benefits include save the attendance data, facilitate the creation of various attendance reports, and make the attendance decision-making process is simple. Moreover, one of the main distinguishing characteristics of their proposed system is that the hardware required is minimal.

K. Ashwin et al. [47] proposed a method to monitor student attendance by using the RFID tags in the student ID cards. When a student enters the classroom, the system records the student's ID and uses Geo-fencing technique to track their current location on the campus.

5.3. IoT-based Assessment of Students

IoT presents some new methods to assess the students. The benefit of incorporating IoT technology in students' assessment is to automate some of tedious instructors' tasks and save time. the automation As a result. of grading tasks will assignments certainly reduce instructors' efforts and allowing them to focus more on enhancing their teaching skills.

K. Mershad et al. [48] introduced the "Students Assessment" application integrated within the proposed LearnSmart LMS [28]. The application uses IoT tools and services to automatically assess and grade students after completing laboratory experiments. The proposed application exploited the power and ability of IoT to enable lab instructors to monitor students' activity while they are conducting experiments, and then automatically evaluate and grade them. The proposed application is one of the first applications that uses IoT and machine learning tools to assessment student in laboratory experiments.

M. Rhoads and B. Stachowiak [49] in their book on the added value of technology in teaching, conclude that educational institutes and invest in better instructors must always understanding the importance of regularly analyzing assessments in a qualitative method instead of only depend on quantitative results and how it can really assist improve learners' performance. and in constantly pursuing advanced and efficient assessments tools.

M. Farhan et al. [50] developed an IoT-based interoperable model to analysis of different students' response to e-Learning by using data collection workflow and an algorithm for attention scoring. This was applied to students attending video lectures comprising an electronic learning component of their studies.

S. Shapsough et al. [51] proposed a mobile/tabletbased assessment system that uses smartphones and tablets to carry out evaluations and allowed real-time reporting. They used the architecture of MQTT publish-subscribe to provide almost assessment services in real-time to various types stakeholders of educational process including educational planners, teachers, principals, and parents. The proposed system collected and assessment data involve sensor delivered information from mobile devices used by students and teachers. The results of the system analyze generated by the analytical engines by the system were directed in real time to the appropriate stakeholders. A preliminary analysis of the proposed system showed promising results in terms of low power consumption in mobile devices and good scalability.

P. Verma et al. [52] presented a system for students to communicate with each other and objects related to the course. They proposed smart computing-based framework is divided into five-layers to facilitate the automation process of student performance assessment in engineering institutions. The daily activity dataset of students

created based on the sensing ability of IoT nodes. They used GPS and RFID sensors to obtain the location of students and objects, and analyzed the activities of students. The collected data is used to evaluate students' performance in the course.

6. IoT Benefits of Administration

Here we will focus on the potential benefits that would be obtained from IoT-based systems for dean, manager or the provost. IoT has the potential to produce significant gains in higher education institutions, especially in the automation of buildings, maintenance systems, energy management, environmental control, access systems to buildings and spaces, large systems of research environments, academic learning systems and safety systems for students, teachers, staff and the public. Many researchers discussed and proposed different IoT-based systems which offer opportunities for admin such as smart monitoring, increase security and safety, institutional energy management and LMS content can be modified.

6.1. IoT-based Eco-System Monitoring

IoT assists in smart monitoring of buildings including HVAC system, lighting, locks and monitoring students. M. Cata [53] proposed a model for a smart university that uses five main categories of sensor devices. According to many applications inside smart universities, they benefited from IoT services, such as: (1) smart parking that monitor vacant parking, thereby avoiding congestion or accidents; (2) smart lighting that can be based on external natural light data collected from sensors in order to automatically reduce classroom lighting, thereby reducing power consumption; (3) smart tracking, by using RFID technology to evaluate any emergency, so as to monitor the goods and equipment inside the university; (4) smart inventory, which can read QR by using a barcode

reader labels to identify any equipment associated with barcodes. Since the health state of students is a key factor in determining the overall academic performance of students, it is essential to obtain high-quality medical services in any educational institution.

With respect to student monitoring, M. Kassab et al. [12] discussed some researches that attempted to monitor students in online laboratory and education settings. Due to the fact that student health state is extremely affect student's overall academic performance, thus granting access to the health care services is important for establishing effective education. A study [54] showed how to use RFID to build eHealth system to monitor students at risk of hypertension as a wearable device. The system focuses on some health information of various students, such as medical history, electrocardiography (ECG) results, prescriptions, blood pressure, and other vital signs by IoT technology.

6.2. IoT-based Energy Management

IoT has been used in energy management to improve energy efficiency for a more sustainable future. This has led to the introduction of Smart Grid, which is a special form of energy management application in the IoT by many governments [55].

In order to provide a healthy atmosphere for learning and teaching, IoT allows the creation of a green campus environment for universities by reducing monitoring and controlling energy, CO2 emissions and water use [56]. With respect to energy saving, a work published in [57] proposed a green campus architecture to manage computers and air conditioning systems.

As a real example, COMFORTSENSE is an energy management system, which implemented in Campus Luigi Einaudi, University of Turin. The project utilizes the benefits of IoT technologies to advance building energy efficiency and comfort [58]. The focus of this system is to consider campus areas and university buildings as a "living lab", and thus collect data through smart phones and wearable devices instead of using traditional data collection ways.

6.3. IoT-based Security and Safety

IOT can improve security and safety. Smart technology can make schools safer. Several academic institutions for instance San Francisco University have combined technologies of IoT to enhance security and safety of campus[2]. For example, wearable wristbands can register entering of students to academic institution premises and classrooms. No more strolling between lessons or during lunch time. This measure also helps prevent unauthorized persons and other intruders from entering the school. In addition, more reliable surveillance cameras are used to monitor corridors and surrounding buildings to ensure the safety of students [3].

A. B. Manduri et al.[4] used the core technologies of edge computing, IOT and cloud computing design the framework of an intelligent emergency alert system. IOT is committed to developing the world smarter and can make a significant contribution to the design of the Framework of Smart Security (SSF) for educational institutions. They tested the proposed system on Cisco Packet Tracer 7. The results showed that the approach used in this system proposed plays an effective role in alerting security not only in institutions of educational but also in other institutions as well.

With the combined technology of IoT and RFID, schools can generate real-time pre-recorded announcements for various emergencies. Educational institutions can immediately close the door in case of panic or emergency. In addition, parents can also track their children's real-time movement, especially the duration of school entry and exit at the same time. At the same time, the staff can ensure that the students are in the classroom they should go to [5]. D. Palma et al. [6] developed an NFC-based classroom access control method to create a realtime classroom control tool. The system of classroom registration is based on a network of a set of connected sensors that collect information about classroom access and display the classroom status on a web-based application and TV panels of university.

7. Challenges with Integration of IoT in Education

IoT technology plays an important role in improving the educational processes. IoT influences on student, instructor and admin, in addition it transforms and improves education methods. Many studies have discussed the benefits of IoT in the educational process, both for the student and for any participant in the education process. On the contrary, there are many challenges associated with applications of IoT in education filed. Some issues are very specific and affect only a small number of people, such as disability, while others are more common instance. adaptation [15]. For to new technologies. According to the analyzed paper, the challenges to education in general that need further research are:

7.1. Security and Privacy

Security requirements have always been an important aspect of education filed. Concerns about security have increased due to the increase in communications and complexity of IoT technology. The "security / privacy" quality concerns are the most discussed because most studies have discussed them [12]. One of the most important security issues facing the education is the exposure of educational institutions to electronic attacks (cyberattacks), which will prevent schools / universities from operating intended. Due as to the

interconnectedness of the large number of devices, Cascade failures may appear, making it difficult to protect simultaneously with all related problems via antenna transmission. Data breaches considered a security challenge. Regarding privacy, whether that monitoring is known or unknown, many devices used in specialized preconfigured IoT will collect different data. But why collect this data, who owns it and where is it going? The legal professions, government agencies that oversee education, and a series of education standards should answer these questions [62].

7.2. Humanization

The ethical role that IoT may play in a person's life is questionable, especially with regard to personal control. IoT applications involve more than just devices that interact with each other. The success of IoT mainly depends on the way of technology, connection rather than the humanization of connection technology. IoT may make people move toward specific habits, thereby reducing people's autonomy, thereby transferring power to companies that focus on gains financial. This means for the system of education that the influential factor is the that controls the tools organizations and techniques used by academic professionals, not the academic professionals themselves. The valid concern is dehumanization in interacting of humans with machines. Numerous studies indicate that the interaction between students face to face will benefit social skills of children, and it will contribute positively to build of character. Hence, going to school is the problem that may arise from the increase in IoT technologies in education as a result of the partial loss of the social aspect [12].

7.3. Scalability

Scalability creates concerns regarding the big data and the cost of implementing IoT in education. The integration of the IoT in education filed is no exception. Incorporating the IoT into education will generate great volume of data. Therefore, the data needs to be managed, analyzed and processed to obtain information and trends. Moreover, there have been several papers focusing on data visualization technology and dashboards. There is still a long way to go to deal with the large amount of data generated in the intelligent education environment, display them correctly and make it easier for students and teachers to understand the data. [26]. With scalability, concerns related to discussion about the cost of IoT technology in education have also become important [30]. Financially, there are immature experiences everywhere [11]. For example, admin, student and teacher training [25], the reduction of public service investment, the current situation of the world economy [1], and the limited resources of universities have prevented the smart campus from becoming a reality [19] The question that arises is how will schools and universities purchase and maintain equipment? The financial obligations this resulting from the shift to an IoT ecosystem focused on education cannot be underestimated. The economic benefits of IoT in education are discussed in the following four selected articles [54].

7.4. Adoption ICT in education

In many developed countries the use of ICTs in education extensively studied, yet in developing countries it is still neglected. In order to help decision-makers take effective measures for its dissemination, researchers and scientists need to examine the factors that affect its accreditation.

7.5. Adapt and Attract Education

Most students use multiple devices and social networks to connect their education to the Internet. More research is needed to understand how students connect to the Internet to increase their learning and application of knowledge. An effective digital curriculum model can guide the current work of using IoT in education. Scientific research aims to improve the existing teaching structure and forms to create personalized and digital curricula in effective manner for ordinary and disabled students to use in informal and formal education: what makes students more interested and attracted is that they go beyond the ideas that are be traditional and the concepts and learning activities to create something novel and meaningful things.

8. Conclusion

IoT improves learning, teaching, and management, and many educational activities. In this paper, a survey of IoT in education activities specifically, to study how this kind of IoT technology with its unique system functions (such as sensing and decision-making) supports and challenges the pedagogical process of all relevant participants (teachers, staff and students). In addition, IoT challenges are showed and explained. As a future work, it is possible to work on enhancing the current work by presenting IoT technologies and tools for each IoT services and opportunities for each participant in the educational process separately and proposed new techniques can solve the challenge of IoT in education.

9. References

- D. D. Ramlowat and B. K. Pattanayak, "Exploring the internet of things (IoT) in education: a review," in Information Systems Design and Intelligent Applications: Springer, 2019, pp. 245-255.
- [2] C.-E. Cornel, "The role of internet of things

for a continuous improvement in education," Hyperion Economic Journal, vol. 2, no. 3, pp. 24-31, 2015.

- [3] E. Oriwoh and M. Conrad, "'Things' in the Internet of Things: towards a definition," International Journal of Internet of Things, vol. 4, no. 1, pp. 1-5, 2015.
- [4] S. Barakat, "Education and the Internet of Everything," Int. Bus. Manag, vol. 10, no. 18, pp. 4301-4303, 2016.
- [5] S. Kumar, T. Sriraksha, and N. Saba, "An IoT based secured smart e-Campus," International Journal of Humanities and Social Science Invention, vol. 6, no. 3, pp. 88-93, 2017.
- [6] I. Bandara and F. Ioras, "The evolving challenges of internet everything: of enhancing student performance and employability higher education," in in INTED2016 10th annual International Technology, Education and Development, 2016, pp. 652-658.
- [7] A. Alalade, J. Ejemeyovwi, E. Ekong, and D. Adeyemo, "Internet of Things as a tool for enhancement of education administration and delivery."
- [8] K. Ashton, "That 'internet of things' thing," RFID journal, vol. 22, no. 7, pp. 97-114, 2009.
- [9] P. Suresh, J. V. Daniel, V. Parthasarathy, and R. Aswathy, "A state of the art review on the Internet of Things (IoT) history, technology and fields of deployment," in 2014 conference International on science engineering and management research (ICSEMR), 2014, pp. 1-8.
- [10] J. Petäjäjärvi, K. Mikhaylov, R. Vuohtoniemi, H. Karvonen, and J. Iinatti, "On the human body communications: wake-up receiver design and channel characterization," EURASIP Journal on Wireless Communications and Networking, vol. 2016, no. 1, p. 179, 2016.
- [11] F. Khelifi, A. Bradai, A. Benslimane, P. Rawat, and M. Atri, "A survey of localization systems in internet of things," Mobile Networks and Applications, vol. 24, no. 3, pp. 761-785, 2019.
- [12] M. Kassab, J. DeFranco, and P. Laplante,

"A systematic literature review on Internet of things in education: Benefits and challenges," Journal of Computer Assisted Learning, vol. 36, no. 2, pp. 115-127, 2020.

- [13] M. Veeramanickam and M. Mohanapriya, "Iot enabled futurus smart campus with effective e-learning: i-campus," GSTF journal of Engineering Technology (JET), vol. 3, no. 4, pp. 8-87, 2016.
- [14] M. Al-Emran, S. I. Malik, and M. N. Al-Kabi, "A Survey of Internet of Things (IoT) in Education: Opportunities and Challenges," in Toward Social Internet of Things (SIoT): Enabling Technologies, Architectures and Applications: Springer, 2020, pp. 197-209.
- [15] A. C. Martín, C. Alario-Hoyos, and C. D. Kloos, "Smart Education: A Review and Future Research Directions," in Multidisciplinary Digital Publishing Institute Proceedings, 2019, vol. 31, no. 1, p. 57.
- [16] S. Pervez, S. ur Rehman, and G. Alandjani, "Role of internet of things (iot) in higher education," in Proceedings of ADVED 2018-4th International Conference on Advances in Education and Social Sciences, 2018, pp. 792-800.
- [17] A. Das, A. Hazari, and R. Karmakar, "IOT In Modern Day Education: A Study," no, vol. 1, pp. 331-336.
- [18] A. Sula, E. Spaho, K. Matsuo, L. Barolli, R. Miho, and F. Xhafa, "An IoT-based system for supporting children with autism spectrum disorder," in 2013 Eighth International Conference on Broadband and Wireless Computing, Communication and Applications, 2013, pp. 282-289.
- [19] R. Mehta. (May 5, 2018). Five Ways the Internet of Things is Changing for Education and Learning. Available: https://customerthink.com/five-ways-the-internetof-things-is-changing-for-education-and-learning/
- [20] A. Sula, E. Spaho, K. Matsuo, L. Barolli, R. Miho, and F. Xhafa, "A smart environment and heuristic diagnostic teaching principlebased system for supporting children with autism during learning," in 2014 28th International Conference on Advanced Information Networking and Applications

Workshops, 2014, pp. 31-36: IEEE.

- [21] A. Uzelac, N. Gligoric, and S. Krco, "A comprehensive study of parameters in physical environment that impact students' focus during lecture using Internet of Things," Computers in Human Behavior, vol. 53, pp. 427-434, 2015.
- [22] S. Meacham, A. Stefanidis, L. Gritt, and K. T. Phalp, "Internet of Things for Education: Facilitating Personalised Education from a University's Perspective," 2018.
- [23] M. Kravčík, C. Ullrich, and C. Igel, "The Potential of the Internet of Things for Supporting Learning and Training in the Digital Age," in Positive Learning in the Age of Information: Springer, 2018, pp. 399-412.
- [24] M. A. Chatti, M. R. Agustiawan, M. Jarke, and M. Specht, "Toward a personal learning environment framework," International Journal of Virtual and Personal Learning Environments (IJVPLE), vol. 1, no. 4, pp. 66-85, 2010.
- [25] A. Labus, M. Milutinovic, Đ. Stepanic, M. Stevanovic, and S. Milinovic, "Wearable computing in e-education," RUO. Revija za Univerzalno Odlicnost, vol. 4, no. 1, p. A39, 2015.
- [26] K. Papadokostaki, S. Panagiotakis, A. Malamos, and K. Vassilakis, "Mobile Learning in the Era of IoT: Is Ubiquitous Learning the Future of Learning?," in Mobile Learning Applications in Early Childhood Education: IGI Global, 2020, pp. 252-280.
- [27] R. Xue, L. Wang, and J. Chen, "Using the IOT to construct ubiquitous learning environment," in 2011 Second International Conference on Mechanic Automation and Control Engineering, 2011, pp. 7878-7880.
- [28] K. Mershad and P. Wakim, "A learning management system enhanced with internet of things applications," Journal of Education and Learning, vol. 7, no. 3, pp. 23-40, 2018.
- [29] M. B. Abbasy and E. V. Quesada, "Predictable influence of IoT (Internet of Things) in the higher education," International Journal of Information and Education Technology, vol. 7, no. 12, pp.

914-920, 2017.

- [30] R. Wan, "Network interactive platform ideological and political education based on Internet technology," in 2016 International Conference on Economy, Management and Education Technology, 2016: Atlantis Press.
- [31] M. S. Satu, S. Roy, F. Akhter, and M. Whaiduzzaman, "IoLT: An IoT Based Collaborative Blended Learning Platform in Higher Education," in 2018 International Conference on Innovation in Engineering and technology (ICIET), 2018, pp. 1-6: IEEE.
- [32] J. Henry, S. Tang, M. Hannenghan, and C. Carter, "A Measure of Student Engagement for Serious Games and IoT," in International Conference on Technologies for E-Learning and Digital Entertainment, 2017, pp. 262-270: Springer.
- [33] R. Garris, R. Ahlers, and J. E. Driskell, "Games, motivation, and learning: A research and practice model," Simulation & gaming, vol. 33, no. 4, pp. 441-467, 2002.
- [34] Z. AjazMoharkan, T. Choudhury, S. C. Gupta, and G. Raj, "Internet of Things and its applications in E-learning," in 2017 3rd International Conference on Computational Intelligence & Communication Technology (CICT), 2017, pp. 1-5.
- [35] I. Asseo, M. Johnson, B. Nilsson, N. Chalapathy, and T. Costello, "The Internet of things: Riding the wave in higher education," EDUCAUSE review, vol. 51, pp. 11-33, 2016.
- [36] A. Alalade, J. Ejemeyovwi, E. Ekong, and A. Adeyemo, "Internet of things as a tool for enhancement of education administration and delivery," International Journal of Mechanical Engineering and Technology, vol. 10, pp. 48-62, 2019.
- [37] J. Vihervaara and T. Alapaholuoma, "Internet of Things: Opportunities for Vocational Education and Training," 2017.
- [38] A. Rytivaara, "Collaborative classroom management in a co-taught primary school classroom," International Journal of Educational Research, vol. 53, pp. 182-191, 2012.
- [39] F. Akter, A. B. Akhi, N. J. Farin, M. M.

Khondoker, and M. G. Saklayen, "IoTSAMS: A novel framework for Internet of Things (IoT) based smart attendance management system," Intelligent Control and Automation, vol. 9, no. 3, pp. 74-84, 2018.

- [40] D. C. L. Max Meyers, ""Can the Internet of Things make education more studentfocused?", 10th Dec 2015.
- [41] S. N. Shruti Ramesh Babu, S. Prabakaran, "Attendance Management Using Automatic Face Tracking System," International Journal of Engineering and Advanced Technology (IJEAT) vol. 8, no. 3S, February 2019.
- [42] Z. Jiang, "Analysis of student activities trajectory and design of attendance management based on internet of things," in 2016 International Conference on Audio, Language and Image Processing (ICALIP), 2016, pp. 600-603.
- [43] P. Tan, H. Wu, P. Li, and H. Xu, "Teaching management system with applications of RFID and IoT technology," Education Sciences, vol. 8, no. 1, p. 26, 2018.
- [44] H. D. Rjeib, N. S. Ali, A. Al Farawn, B. Al-Sadawi, and H. Alsharqi, "Attendance and information system using RFID and webbased application for academic sector," International Journal of Advanced Computer Science and Applications, vol. 9, no. 1, 2018.
- [45] A. Alghamdi and S. Shetty, "Survey toward a smart campus using the internet of things," in 2016 IEEE 4th international conference on future internet of things and cloud (FiCloud), 2016, pp. 235-239: IEEE.
- [46] B. I. Ahmad, "TouchIn: An NFC supported attendance system in a university environment," International Journal of Information and Education Technology, vol. 4, no. 5, p. 448, 2014.
- [47] K. Ashwin, S. Krishnakumar, M. Maheshwari, and A. Perumal, "RFID based student attendance and monitoring system," International Journal of Innovative Research in Computer and Communication Engineering, vol. 3, no. 1, pp. 305-310, 2015.
- [48] K. Mershad, A. Damaj, and A. Hamieh, "Using Internet of Things for Automatic Student Assessment during Laboratory

Experiments," in 2019 IEEE International Smart Cities Conference (ISC2), 2019, pp. 317-323.

- [49] M. Nu-Man and T. Porter, "Igniting Your Teaching with Educational Technology A Resources for New Teachers. Editors: Matt Rhoads & Bonni Stachowiak," ed, 2018.
- [50] M. Farhan et al., "IoT-based students interaction framework using attention-scoring assessment in eLearning," Future Generation Computer Systems, vol. 79, pp. 909-919, 2018.
- [51] S. Shapsough, M. Hassan, S. E. Shapsough, and I. A. Zualkernan, "IoT precision technologies to enhance and response time of mobile-based educational assessments," in 2016 International Conference on Computational Science and Computational Intelligence (CSCI), 2016, pp. 202-205.
- [52] P. Verma, S. K. Sood, and S. Kalra, "Smart computing based student performance evaluation framework for engineering education," Computer Applications in Engineering Education, vol. 25, no. 6, pp. 977-991, 2017.
- [53] M. Caţă, "Smart university, a new concept in the Internet of Things," in 2015 14th RoEduNet International Conference-Networking in Education and Research (RoEduNet NER), 2015, pp. 195-197: IEEE.
- [54] T. Takpor and A. A. Atayero, "Integrating Internet of Things and EHealth solutions for students' healthcare," in Proceedings of the World Congress on Engineering, 2015, vol. 1: World Congress on Engineering, London, UK.
- [55] S. Kim and S. Kim, "A multi-criteria approach toward discovering killer IoT application in Korea," Technological Forecasting and Social Change, vol. 102, pp. 143-155, 2016.
- [56] M. Bagheri and S. H. Movahed, "The effect of the Internet of Things (IoT) on education business model," in 2016 12th International Conference on Signal-Image Technology & Internet-Based Systems (SITIS), 2016, pp. 435-441.

- [57] H.-I. Wang, "Constructing the green campus within the internet of things architecture," International Journal of Distributed Sensor Networks, vol. 10, no. 3, p. 804627, 2014.
- [58] P. V. P. N. Posto, U. DI Professore, and R. DI Ruolo, "università degli studi ditorino."
- [59] Sciforce. (April 9, 2019, March 9,). Internet of Things for the Classroom. Available: https://www.iotforall.com/internetof-things-classroom
- [60] A. Badshah, A. Ghani, M. A. Qureshi, and S. Shamshirband, "Smart security framework for educational institutions using internet of things (IoT)," Comput. Mater. Contin, vol. 61, pp. 81-101, 2019.
- S. Alharbi and S. Drew, "Using the [61] technology acceptance model in understanding academics' behavioural intention to use learning management systems," International Journal of Advanced Computer Science and Applications (IJACSA), vol. 5, no. 1, 2014.
- [62] S. Gul et al., "A survey on role of internet of things in education," IJCSNS, vol. 17, no. 5, p. 159, 2017.

journal homepage: <u>www..ois.sabauni.net</u>



Article

Apracadabra application platform for intelligent and immediate-services based IoT technology

Al-Marhabi Zaid Ali¹, Al-Hamdi Ayeda G², Habeb Abduljlil Abduljlil³

¹Management of information systems department, Faculty of managements and economics, Al-Yemenia University, Sana'a, Yemen

¹Computer science department, Faculty of applied sciences, Hajjah University, Hajjah, Yemen. ²Computer science department, Faculty of applied sciences, Hajjah University, Hajjah, Yemen.

³College of computer science and electronic engineering, Hunan University, Changsha, China

Article info

Keywords:

platform;

IoT:

FRID:

Sensors:

Services)

Article history:

Accepted: May. 2021

Apracadabra application

ASS(Apracadabra Smart

Abstract

Apracadabra application platform visions are to connect multi types of technologies: RFID, sensors, actuators, person's users etc., the opportunity of bring out the IoT technology into reality became possible at the present through the integration of many available technologies, especially in the under development countries (not high infrastructure). The IoT provides many important features, which transformed the internet from communicating with people to the possibility communicating between things "smart devices" to do queries, surveillance and other smart services.

Due to high cost on attaching a sensors or RFID card in some services or providers we propose Apracadabra, Smart Services to contribute on developing societies, would be unable to jump suddenly to the accurate IoT technology concept.

This paper address Apracadabra application platform to be the right solution for under development countries to provide same services as IoT in terms of speed, we can call it the first generation of the IoT. Apracadabra based on attaching an account number or IP address for each user (server - clients), is the services provided by this user will be presented to other interests with the possibility of evaluating those services by the beneficiaries, this platform gives the user's location and all available ways of communicating.

Apracadabra Smart Services (ASS) name came from aladeen Arabic story (Shobic Lubic (شبیك لبیك) or Apracadabra) which means helping others or clients to get whatever they need immediately, it is a mobile devices application running on Android system due to android system widespread and characterized by many features, also we can access Apracadabra through any web browsers.

* Corresponding author: Al-Marhabi Zaid Ali E-mail: marhabi2000@gmail.com

1. Introduction

Internet of Things (IoT) is a novel become very important in modern wireless telecommunications. The concept of IoT based on variety of things or objects such as Radio Frequency Identification(FRID) tags, Sensors, mobile phones, actuators, etc. all this device are able to communicate through unique addressing scheme, which allow them to interact with each other, also provide services to all neighbor's nodes to reach common goals [1].

The First information revolution was the computer in starting from 1940, and internet also counted as second revolution, and now IoT is third revolution of information, Currently, the culture of Internet of Things industry has gained high attention over the world, many of high infrastructure countries look to IoT as new strategic industries and new economic growth engines and sensationally start running IoT in many business field like Amazon.

What is the meaning of unique addressing of objects? the addressing or IP addressing is the way of given ID for each device to allow it to exchange information, but having unique IP is big challenges in this third Generation which called "Semantic oriented", perspective of IoT.

We have three new concepts: things-oriented, internet-oriented and Semantic-oriented technologies and standards, so IoT become the central role in all above mention technologies, as illustrated in Figure 1 [2].

The terms Internet of Things(IoT) derived from Things Oriented for words 'things', thus IoT means Auto-ID [3].

2. Architecture of Apracadabra

The proposed application platform main advantages are the availability to connect many technologies interface, Service Oriented Architecture(SOA) is best software layer architecture proposed for the new growth IoT technology we call it Middle-ware, which consist of five layers, ordered from top to down: application, services composition, service management, object abstraction and the lowest one is the objects itself as illustrated in Figure 2.



Figure 1. The IoT paradigms [2].

Apracadabra application platform visions are to connect multi types of technologies: RFID, sensors [6], actuators, person's users etc., all these can represent a Units to be connected to could internet(if the services providers wants to connect all items using RFID tags and adopts a RFID reader in his stores then the clients can directly communicate with the item through a middleware), in the RFID tags we can insert all information in concern like price, model, type and size about the items, the RFID reader will work as a middleware between the tag and internet then to the far end or clients. In the sensors case: The Base Station(BS) can work as a middleware to help users or client to collect information from different monitoring environments, on the client side the users can search or access the services application interface from android smart phone or even through any web browser such (IE).



Figure 2. Apracadabra application platform architecture [3,5].

3. Application Vision

To understand how the application work let consider a customer driving his car on a road connecting two cities and suddenly got drive failure and perhaps by utilizing his simple experience he can determine the failure type. All what he needs to do is open mobile phone and insert the type of problem he got in his car on Apracadabra application platform interface. it immediately processes his order and guide him to the nearest car mechanic which he is specialist in this type of failure, as well as through the GPS it can determine the mechanic location, distance and provide all contact information about him. The mentioned example is one of the online services provided by the Apracadabra, in searching process this application depends on the distance between the service requester (clientcustomer) and the service provider(saller) to show the results as in Nearest First(NF) concept.

The service provider shop has RFID reader that connects the shop online via the internet to allow the client to search for tools or a particular service. when the client searching about this particular service, through the coming results the costumer can identify the service or item in details and he can buy them through his connected bank account and the Apracadabra uses semantic search as essential benefit as in Figure 2. to show the full relationship between both internet, semantic search and the IoT. Therefore, we conclude that our research is a good example of the desired integration between the three significant techniques and this is one of the most important Apracadabra application platform features.

One of the important aspect of Apracadabra application platform is how can a customer ask for service and how the service provider response and deliver service to him, on other word how/what is the transaction architecture between the two sides (customers- providers)? To answer this question, we find that the appropriate architecture for Apracadabra on the 3rd world countries is to use (customer- internet couldservice provider) because the application developer's future plan to cover different types of services, and it can be extended day by day (whenever new service has been added thus new class will be added too).

The steps of requesting or searching for services as following: a) opening the mobile or any smart phone, which must be connected to internet, b) opening Abracadabra main interface and login in with the customers User name, c) typing the name of the service or products on the textbox then he can choose the class or department which the service belong to, or even without choosing it the application will automatically search on all services classes same as famous searching engines like google. d) after pressing search button the results will be listed directly, we mentioned above that Abracadabra application platform differ from other application searching results which is sorted in Abracadabra on Nearest-First(NF) principle, also the results come according to rate feedback from previous

benefited customers. e) The customer can request one or more product or service (also he can view more detail about such services). f) If the customer wants to contact the service provider he can do that using any social media or even they can use video conferencing.

easily publishing and spreading For the application we plan to link it with some famous application like WhatsApp, WeChat according location user preferring application, we expecting the application will get high popularity in very short time because of its benefits and usability to everyone, in case of willing any user(storemarket- engineer etc..) to advertise any new service he can do it in very easy way though Apracadabra application platform by writing all important information in detail about this service so the application can deliver it to all customers any different ways: a) a new notification bob up in users(who is in interest) in their main page of the application, when he click it he can open new page, in that page he can view all information or data about the promotions, or new services or products etc.. b) The also application had the ability to analyses the data in any (shop- services providers) from transaction record for example to know which customers is interesting to which services, thus the application automatically shows up any news about such service, this is one of data mining aspects, as we mentioned above that one of the most IoT advantages is connecting many technologies together to provide many services and to minimize the cost and efforts.

3.1. Communication types

As we mentioned above Abracadabra application for intelligent services is in the first version, we aim to deploy the fully IoT technology in the 3rd words countries which need such services very urgent, the application also as it have different working architectures with many communicating interfaces it can be also useful in 1st and 2nd word countries, so we can call it a real IoT technology, Apracadabra have communicating types:

- 1. Clients- Service Providers communicating: it means services requesters and providers, or human to human.
- 2. Clients- Services communicating: it means directly connecting a services requesters and products in case of attaching a RFID rag or sensors on the products.

In all cases we can successfully got an excellent results and best services finder based on IoT, whatever these services were buying a goods (online shopping) or an urgent service for customers like searching for care mechanics.

Apracadabra also can directly communicate with the services, products and tools through RFID technology or any collecting data interface, the service provider can set a RFID sticker in the tools or products that service provider need to sell and he can insert and attach all information about this tool inside the Card such as name, model, type, country, price and other concern information for client.

- App back bone: as it's the first version of Apracadabra application its working based on the internet and it use any famous chatting application like Facebook messenger or WhatsApp for connecting the services provider with any client or customers through Apracadabra servers, thus they can communicate and try to get more clarify from services providers, thus our application use a social media just from communicating between two sides (service owner or provider and service requester or client).
- App work flow: Apracadabra application based on U2U architecture and NF (Nearest-First), so when the customer search for any service or products the results will be showed in his screen is based on NF.

3.2. Location detection

The application also has the ability to detect the location for both services and customers through the following:

- When adding new products or new services the person who enter the data can insert the location of his business or market (in case of having fixed location).
- In case of car mechanics or the services provider doesn't have an exact place the application can detect the location of customer or providers based on GPS, thus we can know the distance between them.

4. IoT phase (1)

At first glance, you may see that this application does not use the full methodology of IoT, you may see that this application does not use Internet of Things technology stuff full communicating methodology. However, it contributes and provides a quick and speedy service for those interested in the search for a goods or service and communicate with those who provide this service, thus Apracadabra provides better things or customers communicating all this contribute in fast easy immediate services.

Therefore, this application is very useful in emergency cases, it's based on the number of users i.e. the system working efficiently when it has a large number of users and service providers. Thus, both Apracadabra users works as a server and client at the same time i.e. if the service providers have products can be provided to clients. he need to enter all information about it if client wants the request this product he will search about it, he will get all that information and he can contact with the service provider, in Figure 3. there is chat button allow the two side to chat with each other.

The ASS allows us to also many ways of communication between the clients and service

providers such as direct contact with the instant messaging, audio or video call. And through visual communication clarifies or review the type of service or car failure (as the previous example), or through SMS services.

The Apracadabra application is a Users2Units (U2U) with the opportunity to learn more details about services. This research seeks to utilize IoT system and enables users to find out the location and additional information about the emergency services which they are seeking.

One of the most important success factor on developing this technology and success in order working as real IoT is that the seller enter detailed, accurate information about the available various products, he can also enter additional information about available quantity and price, specifications and the more detailed. Whenever information is accurate the Apracadabra has opportunity to bring more customers, and also saves a lot of effort for those wishing to inquire the current given about the commodity possibilities, we see such an application was considered sufficient, we can call it the initial release phase (1) of the IoT.

Application service types: this application has unlimited service areas so can we call it Unlimited Things Communicating (UTC) thus the database developed to cover all services that we need or want to query or to display to customers so this application can be also help the Hyper Market in easily products advertisement's, but it focuses more on the style of the NFFS concept and its one of the most searching results. This application differs from normal searching sites, its more focused on providing services in the range within a city or smaller than a city (as in the previous example), this does not mean that the application not display the results located in other cities or countries, but as we mentioned earlier it focuses on the principle of Nearest First-Fastest Service (NFFS).

ASS users as mentioned can be a service provider represents many products or services such as (vehicles spare parts with many products and tools) and therefore every item needs to have a unique tag, in FRID case the service provider need it insert all concerned information or characteristics about the item inside the tag.

The other type is also the client (Mechanical Engineer) and thus will need a unique address or user name to represent him, so he can write in detail what kind of failure he has fluent experience to fix and repair it.



Figure 3. Products detail view and customer/ provider communicating.

5. Apracadabra interfaces

Adding new products or new services can be done in two main ways:

- 1. Connecting the products directly through RFID tag or sensor node: this type almost IoT technology, the service provider can just place the RFID tag on all products in his markets and insert all information about the product into the tags.
- 2. Connecting all products which the service providers want offer it to the customers by provider User name or his Account, then under his account he can list all service he offers with correct name and detailed information to allow the customer easily found it when they search. Here we would like to mention the market owner or service providers usually have not enough skills to insert the products information

3. thus he can ask for help from one of application developers or any experts.

This application also provides an immediate advertisement services for all users and customers as fallowing: immediate publicity and advertising process for all interested (if the service provider need to make an announcement about a particular services or any hyper markets make a discount for some products), the application provides notice to all interested in that service and certainly their location, so all clients can see a pop up in his home screen about new products as illustrated in the Figure 4.



Figure 4. Searching result and Apracadabra options.

It's important for any user or client to mention his interests(whatever its products or services), from users interest, thus the application performance will be improved, so when any Hyper Market have new products, promotion or any service provider have new services want to advertise it to the customers(users) they can easily inserts it as new (services or products) as illustrated in Figure 5, when any client open the application in his mobile he can see every new adds as bob up in to top of the screen, this option it's very important especially for small business, it can help them to get more and more customers so their business will grow up faster with cost lower than any other advertising media.



Figure 5. Adding new service or products.

6. Presentation and evaluation

Any user (customer- service providers) have unique account, through this account he can show all services he can offer it to other customers, also he can add new service or new products, thus all this services will be allowed for any customer to view it, also all users have their own page contain all transaction record they already done!

All customer or service benefiter can evaluate this service after they buy it or use it by two ways:

- Through direct evaluation from customers who utilize this service, but we all know most of customers didn't come back to make evaluation for the goods or service they buy thus we can go to second way,
- 2. The application has the ability of limitation all customer who buy or use this service thus from knowledge base the application can evaluate this service. The service which has most positive evaluation rates will get high rank thus it will be in the top of searching results for customers.

This article focus on smart services containing (smart marketing, smart advertisement, easier and

smooth communicating in emergency cases etc.), on another word how this application can help us to find anything we are looking for immediately.

7. Conclusions

Apracadabra application visions are to connect multi types of technologies: RFID, sensors, actuators, person's users etc.., the opportunity of bring out the IoT technology into reality became possible at the

present through the integration of many available technologies, especially in the developing countries, because it doesn't need high infrastructure.

The ASS contributes on enabling users to find out stuff that they are looking for, and provides sufficient and accurate information about products, tools and services they immediate needed with detailed information about it such as location, contact information, and direct contact with providers or even with these stuffs. From the previous section we view the main interface and the application vision, we evaluate the service provided by app and how it can directly link the customers to services which its urgent to him.

In the future we plan to cover different types of services, and it can be extended day by day whenever new service has been added. Thus, new classes will be added too, also we plan to develop the 2nd phase to Apracadabra.

8. References

- [1] D. Giusto, A. Iera, G. Morabito, L. Atzori (Eds.), The Internet of Things, Springer, 2010. ISBN: 978-1-4419-1673-0.
- [2] Shelby, Z. ETSI M2M Standardization, March, 2009. http://zachshelby.org.
- [3] Santucci, G. (September 2009). Internet of the future and internet of things: What is at stake and how are we getting prepared for them? In:

eMatch'99- Future internet workshop, Oslo, Norway, 2009.

- [4] J. Sung, T. Sanchez Lopez, D. Kim, The EPC sensor network for RFID and WSN integration infrastructure, in: Proceedings of IEEE PerComW'07, White Plains, NY, USA, March 2007.
- [5] The EPCglobal Architecture Framework, EPCglobal Final Version 1.3, Approved 19 March 2009, <www.epcglobalinc.org>.
- [6] Alkhalidi Sadam, WangDong and Al-Marhabi Zaid, "Sector-Based Charging Schedule in Rechargeable Wireless Sensor Networks" KSII Transactions on Internet and Information Systems, vol. 11, no. 5, pp. 2310-2345, 2017.
- [7] Mohammed Ali Mohammed MOQBEL, Wangdong, Al-marhabi Zaid Ali, "MIMO Channel Estimation Using the LS and MMSE Algorithm", IOSR Journal of Electronics and Communication Engineering (IOSR-JECE), vol. 12, pp 13-22, 2017.
- [8] Alkhalidi, S.M.; Wang, D.; Al-Marhabi, Z.A. Adopting Sector-Based Replacement (SBR) and Utilizing Air-R to Achieve R-WSN Sustainability. Information 2017.
- [9] K. Sakamura, Challenges in the age of ubiquitous computing: a casestudy of Tengine – an open development platform for embedded systems, in: Proceedings of ICSE'06, Shanghai, China, May 2006.
- [10] Zaid A. Ali Al-Marhabi, LiRen Fa, FanZi Zeng, Ayeda G. Ali Al-Hamdi, "The Design and Evaluation of a Hybrid Compression Technique (HCT) for Wireless Sensor Network", JDCTA: International Journal of Digital Content Technology and its Applications, vol. 5, pp. 201-207, 2011.
- [11] Zaid A. Ali Al-Marhabi, LiRen Fa, FanZi Zeng, Maan Younus Abdullah Alfathi, "HCT Plus based on diminishing WSN Energy Consumption", JDCTA: International Journal

of Digital Content Technology and its Applications, vol. 6, pp. 45 -53, 2012.

- [12] Zaid A. Ali Al-Marhabi, LiRen Fa, FanZi Zeng, Maan Younus Abdullah Alfathi, Alhamidi Radman, "Achieving WSN Performance and Forest Monitoring System with WSC", IJACT: International Journal of Advancements in Computing Technology, vol. 4, pp. 77-84, 2012.
- [13] Wang, M., M., Cao, J., N., Li, J., Das, S., K.: Middleware for Wireless Sensor Networks: A Survey. In: Journal of Computer Science and Technology, vol. 23, pp. 305-326, 2008.
- [14] Henricksen, K., Robinson, R. A Survey of Middleware for Sensor Networks: State-ofthe-Art and Future Directions. In: International Workshop on Middleware for Sensor Networks, Melbourne, Australia, November, pp. 60-65, 2006.
- [15] Sadjadi, S. M., McKinley, P.: A Survey of Adaptive Middleware. Technical Report MSU-CSE- 03-35, Computer Science and Engineering, Michigan State University, East Lansing, Michigan, 2003.
- [16] Atzori, L., Iera, A., Morabito, G.: The Internet of Things: A Survey. In: Computer Networks, vol. 54, pp. 2787-2805, 2010.
- [17] Vision and Challenges for Realizing the Internet of Things, <u>http://ec.europa.eu/</u> information_society/events/shanghai2010/pdf /cerp_iot_clusterbook_2009.pdf, 2010.